Author's response to reviews

Title: INSPIRE: A phase III study of the BLP25 liposome vaccine (L-BLP25) in Asian patients with unresectable stage III non-small cell lung cancer

Authors:

Yi-Long Wu (syylwu@live.cn)
Keunchil Park (kpark@skku.edu)
Ross A Soo (ross.soo@nuhs.edu.sg)
Yan Sun (suny@cisco.org.cn)
Karin Tyroller (karin.tyroller@merck.de)
David Wages (david.wages@emdserono.com)
Guy Ely (guy.ely@merck.de)
James Chih-Hsin Yang (chihyang@ntu.edu.tw)
Tony Mok (tony@clo.cuhk.edu.hk)

Version: 3 Date: 3 October 2011

Author's response to reviews: see over
To the Editor-in-Chief:

Re: INSPIRE manuscript: requested formatting amends
(reference 9182620125454170)

We are pleased to hear of the editorial acceptance of our manuscript, “A phase III study of the BLP25 liposome vaccine (L-BLP25) in Asian patients with unresectable stage III non-small cell lung cancer”.

A point-by-point description of the formatting changes made, as requested, is supplied below:

1. Abstract: the heading 'Conclusions' has been replaced with 'Discussion'
2. Tables: both tables have been reformatted and bullets removed
3. The highlighting has been removed throughout the document
4. Title page: The two Merck emails were reformatted in line with the following: IJKL@corporation.com
5. Title page: The word counts have been removed from the title page
6. ‘Acknowledgements' section: This section has been moved so that it follows the 'Author contributions' statement

We hope that the revised manuscript is now suitable for publication in BMC Cancer. Thank you for your time, we look forward to hearing from you.

Best wishes,

Professor Tony Mok

State Key Laboratory of Southern China
The Chinese University of Hong Kong
Sir YK Pau Cancer Center
Prince of Wales Hospital
Hong Kong, China

On behalf of the INSPIRE study group:
Yi-Long Wu, Keunchil Park, Ross A Soo, Yan Sun, Karin Tyroller, David Wages, Guy Ely and James Chih-Hsin Yang